Author's response to reviews

Title: Evaluation of pulse-oximetry oxygen saturation taken through skin protective covering

Authors:

Jyotsna James (jyotsnajames@yahoo.com)
Lokesh Tiwari (lokeshdoc@yahoo.com)
Pramod K Upadhyay (pkumar@nii.res.in)
Vishnubhatla Sreenivas (sreenivas_vishnu@yahoo.com)
Vikas Bhamhani (vikasbhamhani@hotmail.com)
Jacob Puliyel (puliyl@vsnl.com)

Version: 7  Date: 1 May 2006

Author's response to reviews:

Dear Dr Robin Stewart
Thank you for help with the formatting.
I have added an Acknowledgement section after the text but before 'Authors contributions'.
I hope the article is now acceptable
Sincerely
Jacob Puliyel